Author's response to reviews

Title: Socio-Epidemiological Features of 2002 Plague Outbreak in Himachal Pradesh, India

Authors:

Sonu Goel (sonugoel007@gmail.com)
Harvinder Kaur (harvinderkaur315@gmail.com)
A K Gupta (anil1031@hotmail.com)
Umesh S Chauhan (uschauhan78@gmail.com)
Amarjeet Singh (amarminhas56@rediffmail.com)

Version: 5
Date: 20 February 2014

Author's response to reviews: see over
Subject: Revised manuscript 1891864883917771
Socio-Epidemiological Features of 2002 Plague Outbreak in Himachal Pradesh, India

Dear Editor,

We thank the reviewers for prompt response on the article. The point-wise comments are addressed below:

Comment- We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

Response- The copyediting of the paper for written English has now been done by Prof. Deepti Gupta, PhD (English language teaching) and Professor in English department of Panjab University, Chandigarh.

It is kindly requested to acknowledge the receipt of the comments and revised article.

Best Regards

Yours Sincerely,

Dr Sonu Goel

(Corresponding Author)